- 1 How high is high enough? A multi-million-member ensemble analysis of future climate scenarios and
- 2 their relevance
- Marcus C Sarofim^a*, Christopher J Smith^{b,c,d}, Parker Malek^e, Erin E McDuffie^a, Corinne A Hartin^a, Claire
 Lay^e
- 5
- 6 Author Affiliations:
- 7 a: US Environmental Protection Agency, Washington, DC, 20460, USA
- 8 b: Met Office Hadley Centre, Exeter, EX1 3PB, UK
- 9 c: School of Earth and Environment, University of Leeds, LS2 9JT, UK
- 10 d: International Institute for Applied Systems Analysis (IIASA), 2361 Laxenburg, Austria
- 11 e: Abt Associates, Boulder, CO, 80302, USA
- 12 *: Marcus C Sarofim
- 13 US Environmental Protection Agency
- 14 Climate Change Division
- 15 Mail Code 6207A
- 16 1200 Pennsylvania Ave NW
- 17 Washington, DC, 20460, USA
- 18 202-343-9993
- 19 <u>Sarofim.marcus@epa.gov</u>
- 20
- 21 Preprint server: this manuscript is a non-peer reviewed preprint submitted to EarthArXiv
- 22 Peer review journal submission: this manuscript is intended for submission to the Proceedings of the
- 23 National Academy of Sciences of the United States
- 24
- 25 Classification:
- 26 Social Sciences, Environmental Sciences
- 27 Physical Sciences, Environmental Sciences
- 28
- 29 Keywords:
- 30 Climate scenarios, probability, RCP8.5, risk

31 Abstract:

- 32 Assessments of high-forcing climate scenarios provide unique insight into possible high-risk climate
- change impacts in the 21st century and beyond. Given rapid and ongoing societal changes (e.g.,
- 34 population growth, energy demand, technology, etc.), debates are increasing on the continued
- relevance of these 'high-forcing' scenarios, such as those designed to reach 8.5 W/m² by the end of the
- 36 century (e.g., RCP8.5 and SSP5-8.5). Here, we determine an updated probability of exceeding 8.5 W/m²
- by 2100 by generating a multi-million-member ensemble from a newly developed database of
- 38 probabilistic greenhouse gas (GHG) emissions scenarios, augmented here with improved
- 39 characterization of aerosols and minor GHGs, coupled with the newest version of a reduced complexity
- 40 climate model, with parameter distributions calibrated to match the IPCC 6th Assessment Report. The
- 41 probability of exceeding 8.5 W/m² this century is estimated to be less than 1%, however, we discuss
- 42 important rationales for the continued use of 8.5 W/m² scenarios, including for generating high
- 43 temperature scenarios for use in calibrating damage functions, characterizing climate in the 22nd century
- 44 (probability of exceeding 8.5 W/m² increases to ~7% by 2150), and assessing low-probability/high-
- 45 impact futures..

46

47 Significance Statement:

48 Controversy exists regarding the relevance of 8.5 W/m² high-end climate scenarios. We calculate the

- 49 likelihood of these scenarios. We show a less than 1% probability of exceeding 8.5 W/m² by the end of
- 50 the century but also discuss why these scenarios are still useful for anchoring the high end of damage
- 51 functions, for providing analogues for climates after the year 2100, and for investigating the outcomes
- 52 of low probability but high impact futures.
- 53

54 <u>Introduction</u>:

- 55 Representation Concentration Pathway (or RCP) 8.5 was the highest of four concentration scenarios
- 56 (RCP2.6, 4.5, 6.0, and 8.5) developed as inputs for the CMIP5 modeling exercise (Moss et al. 2010, van
- 57 Vuuren et al. 2011). The numerical component of the scenario name refers to the increase in radiative
- 58 forcing between pre-industrial times and 2100 in the scenario in units of watts per square meter
- 59 (W/m²). The RCP scenarios were used in climate models to inform the IPCC Fifth Assessment Report
- 60 (AR5) and the US National Climate Assessment (NCA4), as well as innumerable research papers. While
- 61 NCA4 used the full range of RCP scenarios, the assessment focused most closely on RCP4.5 and RCP8.5,
- 62 which they considered as low-end and high-end scenarios, to be consistent with the range of scenarios
- 63 used in previous assessments (USGCRP, 2015). These two scenarios were also the most heavily
- 64 represented in the existing literature at the time. More recently, a new set of Shared Socioeconomic
- Pathway (SSP) scenarios were developed for the CMIP6 modeling exercise (O'Neill et al. 2016, Riahi et
- al. 2017), which were intended to more fully integrate changes in socioeconomics with changes in
- 67 radiative forcing, and included new radiative forcing endpoints of 1.9, 3.4, and 7.0 W/m², peak and
- 68 decline scenarios, and alternate socioeconomic assumptions (SSP1 through 5) for many of the scenarios.
- The SSP2-4.5 and SSP5-8.5 scenarios are considered the closest analogues to RCP4.5 and 8.5, which may
- account for why these two scenarios are similarly highly represented in the literature.

- 71 RCP8.5 has been characterized as a scenario that assumes high population growth, low income growth,
- and modest improvements in energy intensity over the next century, leading to high energy demand
- and GHG emissions (Riahi et al. 2011, Moss et al. 2010). The research community originally developed
- the four scenarios as "plausible futures", with RCP8.5 roughly representing the upper 10th percentile of
- radiative forcing from scenarios in the published literature at the time of development in 2007 (Moss et
- al. 2010). However, this is not a rigorous method for determining probability of exceedance of a
- 77 scenario. Because no probabilities were assigned to these scenarios from the literature, this 10%
- 78 statistic may bear little resemblance to an actual probability of exceedance: the distribution includes
- both policy and non-policy scenarios leading to an underestimate of BAU emissions; the majority of the
 scenarios are likely to be designed as "best-estimates" which will underweight the tails; and carbon
- scenarios are included as best estimates which will ander weight the tans, and
 cycle uncertainties may not have been included in all the studies, again underweighting tails.
- 82 Due in part to the use of RCP8.5 in the NCA4 and other assessments, as well as the continued use and 83 interest in SSP5-8.5, questions regarding the plausibility (e.g., Christensen et al. 2018, Schwalm et al. 84 2020, or Kemp et al. 2022) or implausibility (e.g., Hausfather and Peters, 2020, Burgess et al. 2020, Skea 85 et al. 2021, Srikrishnan et al. 2022, or Pielke et al. 2022) of 8.5 W/m² scenarios have steadily increased. 86 Additionally, the recent IPCC AR6 report stated that "the likelihood of high-emissions scenarios such as 87 RCP8.5 or SSP5-8.5 is considered low in light of recent developments in the energy sector" (Chen et al. 88 2021). Importantly, while most of these evaluations of RCP8.5 or SSP5-8.5 have focused on the 89 plausibility or implausibility of the fossil fuel usage and/or emissions pathways in the scenario, it is 90 important to recognize that when these scenarios are used as input to global climate models (GCMs), 91 they are primarily used as concentration pathways rather than emission scenarios (several of the studies 92 previously referenced do discuss the potential importance of carbon cycle uncertainties, though without 93 analyzing the implications of including those uncertainties – e.g., Hausfather and Peters 2020, Chen et 94 al. 2021, and Srikrishnan et al. 2022). Huard et al. 2022 has also highlighted this issue of evaluating 95 scenarios based on their emissions versus their concentrations. Radiative forcing is a simplifying 96 aggregate metric to consider rather than comparing individual greenhouse gas concentrations. This 97 then leads to two related questions: 1) given the latest available societal projections, what is the 98 likelihood of exceeding 8.5 W/m² by the end of the century, and 2) what is the appropriate likelihood for
- 99 the high-end scenario to include in scientific assessments? This paper addresses those questions.
- 100 To answer these questions, this paper has several primary aims. First, we leverage recent advances in 101 probabilistic emission projections (RFF-SPs, Rennert et al. 2022a, Rennert et al. 2022b), and coupled 102 these with a state-of-the-art simple climate model (FaIR version 2.1, Leach et al. 2021), augmented with 103 improved methods to account for emissions of aerosols and minor GHGs (based on Lamboll et al. 2020), 104 to generate a multi-million-member ensemble to derive an updated probability density function for end 105 of century radiative forcing. Second, we use the forcing probability density function from this ensemble 106 of equally probable emissions scenarios and climate parameter combinations to determine the 107 likelihood of exceeding 8.5 W/m² in 2100 (or 2150). Lastly, we discuss considerations for the appropriate
- 108 choice of likelihood for an upper bound, high forcing, scenario depending on the relevant application.
- 109

110 <u>Results</u>:

111 We first compare emission levels across the RFF-SP scenarios to those in the SSP scenarios (Figure 1) by

112 calculating the number of RFF-SP scenarios where CO₂ emissions in 2100 exceed those in six SSP

- scenarios, and then divide by the 10,000 total scenarios to come up with a probability of exceedance.
- 114 We also repeat this calculation with GWP-weighted emissions to account for the role of non-CO₂ GHGs.
- 115 Table 1 shows the results of these calculations, with the 2100 emissions from each SSP scenario included
- 116 in the left column for comparison.
- 117 Figure 1: Mean and the 95% and 5% bounds for CO₂ emissions from the RFF-SP scenarios, as well as CO₂
- emissions from six SSP-RCPs, from 2000-2300



- 119 120
- 121 Table 1: Probability that CO₂ emissions in 2100 (absolute & GWP-weighted) in the 10,000 RFF-SP
- scenarios exceed those in six of the SSP scenarios. Global warming potentials (GWPs) are from IPCC AR4.

Comparison scenario	CO ₂ -only	GWP weighted	Cumulative CO ₂ by
End-of-century emissions			2100 (GtC)
SSP5-8.5	0.030%	0.020%	3078
126 GtCO ₂			
142 GtCO ₂ e			
SSP3-7.0	1.1%	0.46%	2813
82.7 GtCO ₂			
108 GtCO ₂ e			
SSP4-6.0	41.8%	36.3%	2585
21.9 GtCO ₂			
39.0 GtCO ₂ e			
SSP2-4.5	66.3%	71.8%	2523
9.68 GtCO ₂			
19.7 GtCO ₂ e			
SSP4-3.4	100%	100%	2353
-14.8 GtCO ₂			
-0.921 GtCO ₂ e			
SSP1-2.6	100%	100%	2337
-8.62 GtCO ₂			

-3.07 GtCO ₂ e	-3.07 GtCO ₂ e			
---------------------------	---------------------------	--	--	--

The RFF-SP emissions are then used as inputs to the FaIR model. Each of the emissions scenarios is run with all 1001 FaIR parameter combinations, for 10,010,000 total output scenarios. Figure 2 shows the radiative forcing from the SSP scenarios (run in FaIR with the same calibration) overlayed on top of the 90% bounds of this analysis. Table 1 shows the percentage of the runs from this analysis that exceed each forcing threshold in 2100 and then again in 2150.

129

- 130 Figure 2: Mean and the 95% and 5% bounds for total radiative forcing from the RFF-SPs scenarios run
- through FaIR 2.1 with the 1001 parameter sets, as well as total radiative forcing from six SSP-RCP
- emission scenarios run through FaIR 2.1.



133

134 Table 2: Probability that total radiative forcing from the RFF-SP scenarios combined with FaIR climate

uncertainty exceeds each SSP radiative forcing threshold in 2100, 2150, or 2300

Radiative forcing threshold (W/m ²)	Year: 2100	Year: 2150	Year: 2300
8.5	0.53%	6.9%	20%
7.0	7.2%	22%	32%
6.0	25%	37%	43%
4.5	68%	65%	62%
3.4	92%	85%	77%
2.6	99%	95%	86%

136

- 137 To assess the relative contributions of emissions uncertainty and climate parameter uncertainty, we also
- produce Figure 3, in the tradition of Hawkins and Sutton (2009) and Lehner et al. (2020). Figure 3 shows

- a timeseries of the range of radiative forcing and temperatures (an output from FaIR) resulting from the
- 140 1001 FaIR runs around the emissions scenario with median forcing (climate uncertainty), the 10,000 RFF-
- 141 SP runs with a single FaIR parameter set (chosen as having the median climate sensitivity; emissions
- 142 uncertainty), and the combination of the two (total uncertainty), along with the fractional contribution
- of each. Note that climate variability is included in all analyses, such that summing the individual
- 144 uncertainty contributions is expected to exceed the actual total uncertainty.
- 145
- 146 Figure 3: Annual radiative forcing (a,b) and temperature (c,d) projections, showing the 5th to 95th %
- 147 bounds from emissions uncertainty, climate parameter uncertainty, and total combined uncertainty.
- 148 Note that internal climate variability contributes to all three uncertainty distributions.
- 149
- 150





153 Discussion

- 154 The relevance of a particular future scenario depends on the application, purpose, and audience of a
- 155 particular analysis or assessment. However, the relevance also depends on the probability of exceeding
- the metric of interest, whether it be a given quantity of emissions, concentration of CO₂ or CO₂-
- 157 equivalent, total radiative forcing, global temperature, or some other metric. Despite the low probability
- of exceeding 8.5W/m² by the end of the century (Table 2), there are at least three applications where it
- remains crucial to consider these high-forcing, low-probability scenarios: 1) development of damage
- 160 functions or "by-degree" climate impact analyses, 2) as an analogue for post-2100 climate change, and
- 161 3) consideration of low probability but high impact possibilities.
- 162 For the first application, "by-degree" climate impact analyses are mostly scenario agnostic (Sarofim et al.
- 163 2021). However, the higher the emissions driving the analysis, the larger a temperature (or other
- 164 climate parameter) range that will be covered, making a high-end, but plausible, future desirable. As
- damage functions are essentially a mathematical fit to a "by-degree" approach, the anchor for the high
- 166 end of the damage function should be a low probability scenario (note that it can also be important to
- 167 have a similar anchor for the low-end in order to explore marginal benefits of additional mitigation even
- 168 when a stringent policy future is used as the reference). For example, Hartin et al. (2023) uses piecewise

linear extrapolation between data points to assess climate change damages across a given range oftemperature scenarios.

171 The second application regarding post-2100 analogues is relevant because most GCM scenario 172 projections end in 2100 (O'Neill et al. 2016), except for a few GCMs which were run with extended 173 scenarios. Therefore, the majority of the impacts literature is based on scenarios that end in 2100. 174 However, the real world will continue past that date. For this reason, we argue that climate projections 175 for 2100 under high forcing scenarios can be used as analogues for climates that may not be seen until 176 much later. This use of climate projections for one year as an analogue for climate impacts in a different 177 year is the fundamental concept behind the "temperature slice" or "global warming level" approach 178 (e.g., Sarofim et al. 2021, Schleussner et al. 2016). Given this application, we have calculated the 179 probability density function of radiative forcing in 2150 to show that the probability of exceeding 8.5 180 W/m2 reaches almost 7% due to a combination of potential continued emissions as well as climate 181 feedbacks. While the 0.5% probability of exceeding 8.5 W/m2 is below the 1% relevance threshold we 182 discuss below, it is difficult to argue policy irrelevance with a probability of exceedance of greater than 183 5%, even if the date is further in the future. While we have chosen 2150 to illustrate this point, we will 184 note that projections developed for the social cost of greenhouse gas extend through 2300 (Rennert et

al. 2022), and the exceedance probability of 8.5 W/m2 is 20% by the end of this period.

186 For the third application, low probability but high impact events are often of relevance to policymakers. 187 As discussed further in the following paragraphs, the precise probability of exceedance becomes 188 relevant in assessing risk in these contexts. This discussion of probability is also relevant to the choice of 189 the high-end anchor for damage functions or by-degree analyses and for deciding the hottest post-2100 190 potential climate for which to simulate an analogue. The question then becomes how likely does a 191 scenario have to be to be relevant? On the small end, Ansolabehere et al. (2003) recommended an 192 annual risk of 1 in a million for a large, early release of radioactivity from a nuclear plant. On the larger 193 end, few people would carry an umbrella with them if the probability of rain was less than 10%. 194 However, these examples do not provide good context for the appropriate choice of likelihood for a 195 high-end climate scenario. Many potential examples involve repeated probabilities for individual events 196 that can happen at multiple locations (e.g., the nuclear example, or the choice of the 1 in 100 year 197 flooding event for floodplain analysis). In contrast, climate change involves a global, accumulating 198 problem where the probability of a high-end scenario is a function of socioeconomic assumptions and 199 uncertainty in basic scientific parameters. We choose 1% as a policy relevant probability for a high-end 200 scenario for generalized climate analysis purposes, though recognizing that others could choose a 201 different threshold.

202 Several considerations are relevant for these three applications. There is the choice to focus on

203 probability of exceeding radiative forcing rather than emissions. There are challenges involved in

- 204 characterizing uncertainty, particularly for low probability tails of distributions. And there is the
- 205 recognition that even if there are good reasons to include a high scenario in an analysis, that scenario is
- 206 not necessarily the most relevant scenario.
- 207 Because CMIP GCMs and ESMs are generally driven by concentration scenarios, not emission scenarios,
- it is the probability of exceeding those concentrations which is most relevant. Radiative forcing is the
- 209 best available metric to use to represent the aggregate climate impact of a concentration pathway,
- 210 though it has some drawbacks. For example, different GCMs can have different radiative physics, such

- 211 that different GCMs run with the same set of GHG and aerosol concentrations may result in different
- levels of forcing and two emission scenarios with the same forcing in one GCM may diverge when run in
- a second. Relevantly, while the IIASA database for scenarios suggests that RCP8.5 reaches 8.4 W/m2 in
- 214 2100 and SSP5-8.5 reaches 8.7 W/m2, more recent simple climate models incorporating updated
- 215 knowledge in Earth system processes yield higher radiative forcings from the same set of concentrations
- 216 (Byers et al. 2022). The most relevant forcing is the one embodied within the GCM calculations.
- 217 Fredriksen et al. (2023) show that GCM-derived forcing is markedly higher for SSP5-8.5 in CMIP6 models
- than for RCP8.5 in CMIP5 models. Tebaldi et al (2021) estimate that most GCMs reach 8.9 W/m² by 2100
- for SSP5-8.5. Wyser et al. estimated 9.2 W/m² in the EC-Earth3-Veg GCM for SSP5-8.5, but only 7.8
- 220 W/m² for RCP8.5. However, here we focus on radiative forcing as the best available metric, even while
- 221 recognizing its imperfections.
- 222 In addition, when looking at low probability outcomes, considerations of additional sources of unbiased
- 223 (no more likely to be above the reference parameters than below) uncertainty will always increase the
- 224 probability of exceeding those outcomes. For example, while the chance of the world having higher
- forcing than the SSP5-8.5 scenario in 2100 is a low probability outcome, it is more likely than the chance
- of exceeding SSP5-8.5 emissions. In this case, the probability of exceeding the emissions in the SSP5-8.5
- scenario in 2100 when taking the RFF-SP distribution as truth is substantially lower than 1%. Even the
- probability of exceeding SSP3-7.0 emissions by 2100 is on the order of 1%. However, when we move to
- radiative forcing, the probabilities become substantially larger: 0.5% for exceeding 8.5 W/m², and 7% for
- 230 exceeding 7 W/m². A key contributor to this increase in the probability of exceedance is the inclusion of
- additional uncertainties, in particular carbon cycle and methane feedbacks, and methane chemistry
- uncertainties. Because there are more emissions scenarios that are just short of SSP5-8.5 than just larger
- than SSP5-8.5, changing parameter draws would be more likely to push a concentration pathway over
- the SSP5-8.5 threshold than to pull a pathway under it.
- 235 Figure 3 shows that climate parameter uncertainty is the driver of near-term uncertainty in both
- radiative forcing and temperature. Our analysis shows that while climate parameter uncertainty and
- 237 emissions uncertainty become equal contributors to total forcing uncertainty by 2050, climate
- parameter uncertainty exceeds emissions uncertainty almost through 2100 for temperature uncertainty.
- 239 This is a result of there being a more limited set of uncertain climate parameters that impact forcing
- 240 (aerosol forcing, carbon cycle, and methane feedback uncertainties being some of the more important),
- and a more expansive set of climate parameters that impact temperature (climate sensitivity and ocean
- heat uptake being key). While most of this paper discusses radiative forcing as it is the key input to
- climate models, the presentation of temperature uncertainty is important as it is the actual driver of
- impacts that are relevant to human and ecological experiences. Two key advances over the previous
- Hawkins and Sutton (2009) and Lehner et el. (2020) are the use of fully probabilistic emission and
- climate parameter uncertainty inputs for this analysis (rather than, for example, assuming that the RCPs
- 247 are all equally likely).
- Of course, projecting even to 2100, less than 80 years into the future, is a challenging task, particularly
- 249 for socioeconomic conditions that are key factors in determining future emissions. Barron (2018)
- 250 discusses some of the challenges in projecting future penetration of low-carbon technologies using
- 251 climate policy models. While the RFF-SP scenarios may be the best probabilistic emissions forecast
- available, these probabilities must be taken with some skepticism, particularly in the tails (such as the
- 253 1% threshold that we focus on). The RFF-SPs rely primarily on expert elicitation, but there are other

254 approaches. Liu and Raftery (2021) use a Bayesian approach to drive a Markov chain Monte Carlo for 255 developing emission scenarios. Morris et al. (2022) apply a complex computable general equilibrium 256 model driven with uncertain input parameters whose distributions are defined by a combination of 257 expert elicitation and statistical analysis of historical data. All these projections seem to be reasonably 258 well aligned. The definition of "business as usual" can also be a challenging one – for the RFF-SP 259 scenarios, the experts were instructed to provide economic growth estimates absent the effects of 260 future climate change or additional policy efforts to reduce emissions, but several of the experts 261 acknowledged that their final probability distributions did include some internalization of future climate 262 policies and impacts. Moreover, experts were instructed to include views about the evolution of future 263 policy for projecting changes in technology, fuel use, and other socioeconomic conditions relevant to 264 emission factors.

- 265 While there is more confidence in the climate uncertainty probabilities as expressed in the 1001 FaIR
- 266 ensemble members compared with the socioeconomic uncertainties driving emission projections, there
- are also some limitations regarding these physical parameter definitions. In particular, any feedbacks
- that don't exist in either the historical record or the climate models against which FaIR was calibrated
- will not be captured by this analysis. Prior distributions for FaIR parameters were informed by CMIP6
- 270 GCMs, and the potential for the uncertainty space to be under-sampled exists. Furthermore, the
- 271 permafrost feedback is not included in FaIR, which has the potential to contribute substantially to
- 272 radiative forcing and warming through release of additional CO₂ and CH₄.
- 273 While our analysis shows that 8.5 W/m² scenarios are appropriate for use in creating damage functions,
- and important to include in the next round of CMIP analyses, it is important to reiterate that 8.5 W/m² is
- still an unlikely future outcome even through the 22nd century. We are arguing here for the inclusion of
- 8.5 W/m² scenarios to anchor the high end of possible outcomes or to develop damage functions, but
 not necessarily as the most relevant future. One pertinent example is the use of a future scenario for
- not necessarily as the most relevant future. One pertinent example is the use of a future scenario for
 use in adaptation decisions. There are often non-negligible costs associated with over-adaptation: e.g.,
- for sea level rise, there are expenses involved in building higher sea walls than would be optimal from a
- cost-benefit analysis, and in some cases, there could be unnecessary abandonment of properties based
- 281 on over-reliance on pessimistic forecasts. While it may be appropriate to use a scenario that is higher
- than the median or mean scenario, something like the 25% likely 6 W/m² scenario might be better
- suited than the 1% likely 8.5 W/m² scenario. However, this does not mean an 8.5 W/m² scenario can't
- 284 be relevant: the use of a by-degree approach allows for the GCM output of such a scenario to be used as
- analogues for lower temperatures whose probabilities are calculated by the use of reduced complexity
- 286 models (such as FaIR and the RFF-SPs).
- 287 One noteworthy issue is that the estimated likelihood of high forcing scenarios has declined over time.
- 288 While, as discussed in the introduction, there was no formal probability assigned to the 8.5 W/m²
- 289 scenario though it was chosen to represent the upper 10% of existing scenarios (Moss et al., 2010),
- 290 Webster et al. (2009) combined probabilistic estimates of emissions and climate parameters to estimate
- a 90% upper bound of 9.8 W/m² relative to 1990 (or about 11.6 W/m² relative to pre-industrial).
- 292 Therefore, over the past 14 years, the probability of exceeding 8.5 W/m2 by the end of the century has
- 293 dropped from double digits to below one percent. While some of this reduction of the high end of
- 294 possible forcing futures may be due to better information about climate uncertainty, it is likely that the
- 295 majority is due to reductions in estimated future emissions, due to a combination of technological

progress and policy action. This is good news, and the trend towards lower probability of high forcingscenarios will hopefully continue in the future.

298

299 <u>Conclusions</u>

Here we use a 10 million-member-ensemble to analyze the probability of exceeding 8.5 W/m² in 2100,

2150, and 2300. Our results show that while the probability of exceeding this threshold in 2100 is less

than 1%, it reaches 7% by 2150, and 20% by 2300. Because many climate analyses to date have focused

303 on the current century, a scenario that reaches 8.5 W/m² in 2100 continues to have substantial

304 relevance for serving as an analogue for post-2100 climates, as well as for driving impacts models to

develop damage functions that can be useful through the year 2300.

Therefore, it is important that the CMIP community continue to include an 8.5 W/m² scenario in the

family of scenarios for CMIP7. In addition, if an impact modeler is only going to use a single scenario, 8.5

308 W/m² is the appropriate scenario to use – not because it is the most likely, but because it can serve to

inform a by-degree approach which can be coupled with a reduced complexity model to estimate the

timing and probability of reaching future forcing and temperature thresholds, enabling the discussion of

almost any possible future including years past 2100 (with a possible exception of peak-and-decline

312 scenarios). However, if it becomes standard in the future to run impacts analyses through later years

313 (such as 2150), it might be more appropriate to choose a high-end scenario which reaches 8.5 W/m2 in

that later year (rather than 2100).

315 While our analysis suggests that 8.5 W/m2 continues to be sufficiently likely to be relevant based on

316 projections founded on today's best estimates, we hope that as zero carbon technologies continue to

317 become more competitive, and mitigation efforts continue to increase, that soon the 8.5 W/m2 scenario

318 might drop enough in likelihood that our arguments for continuing to include it would no longer hold.

319

320 <u>Methods</u>

We quantify the likelihood of exceeding 8.5 W/m² by the end of the century by developing a probability

density function, based on two components: 1) probabilistic projections of greenhouse gas (GHG) and

aerosol emissions and 2) coupling these with a simple climate model that accounts for additional

324 uncertainty in climate parameters. To derive emission projections, we start with the 10,000 global GHG

emission scenarios from the Resources for the Future Socioeconomic Projections (RFF-SP) dataset,

which provides global CO₂, CH₄ and N₂O emissions from 2020 through 2300 (Rennert et al. 2022a). While

327 the original RFF-SP scenarios assumed that additional short-lived climate forcers (tropospheric ozone

and aerosol precursors) and other greenhouse gases followed the SSP2-4.5 scenario, here we infill the

329 RFF-SP GHG scenarios using the *silicone* tool (Lamboll et al. 2020). Infilling uses a database of existing

integrated assessment model (IAM) GHG and air pollutant emissions scenarios and uses relationships

between a lead species (in our case CO₂ emissions from fossil fuels and industry) and those that are not
 present in the RFF-SPs. The logic behind this is that emissions of CO₂ tend to be well-correlated with

emissions of non-CO₂ species over time (Rogelj et al. 2014) as many species are co-emitted with CO_2 .

333 Where they are not, high emissions scenarios tend to be indicative of high economic activity, low levels

of climate and air quality policy, or both (and vice versa for low emissions scenarios). In this analysis, we

- use the IPCC Sixth Assessment Report (AR6) Working Group 3 (WG3) infiller database (Kikstra et al.
- 337 2022; Byers et al. 2022). The emissions in this database have been harmonized (Gidden et al. 2018) to
- ensure a smooth transition from historically reported emissions values to future scenario projections. As
- the RFF-SP dataset only provides total CO₂ emissions this contribution needs to be split into fossil fuel
- and industrial (FFI) emissions and land use (AFOLU) emissions to run FaIR and perform the infilling. This
- 341 was performed using the silicone toolbox with the "time-dependent ratio" method, mapping FFI and
- AFOLU emissions to total CO₂ emissions from the AR6 infiller database using the average ratio of AFOLU
- and FFI to total CO_2 across the database (Lamboll et al. 2020).
- 344 We replicate the infilling methods of the IPCC Working Group 3 report (Kikstra et al. 2022; Riahi et al.
- 2022) as closely as possible. For example, short-lived climate forcers (SO₂, BC, OC, NH₃, NO_x, CO and
- VOCs) are infilled using the "quantile rolling windows" (QRW) method of silicone. We use CO₂ FFI as the
- lead species and the AR6 infiller database. QRW maps the lead specie (CO₂ FFI) to a specified quantile of
- emissions of the desired pollutant from the infiller database over time. We use the 50th percentile for
- each short-lived forcer. This preserves relationships of co-emittance between high and low emissions
- scenarios (e.g. high CO₂ FFI likely implies high SO₂ emissions in the absence of air quality improvements)
- but does not further differentiate between scenarios (e.g. high CO₂ FFI emissions could occur with low
- 352 SO₂ emissions if air quality improvements occur but emissions mitigation does not).
- All halogenated GHG emissions are infilled using the "RMS Closest" method of silicone. As suggested, this selects the emissions of the desired greenhouse gas from the AR6 scenario based on the scenario
- 355 that minimizes the root-mean-square error between the time series of (reconstructed) CO₂ FFI emissions
- 356 from the RFF-SP scenario and the AR6 scenario. For some of the more major hydrofluorocarbons (HFCs)
- and perfluorocarbons (PFCs) that are individually modelled by several IAM groups, the AR6 infiller
- 358 database is used. In contrast, emissions for more minor gases and all ozone-depleting substances are
- not generally available in IAMs and therefore the set of eight SSP scenarios that drive the CMIP6 Earth
- 360 System models and were prepared on global annual mean emissions timeseries for the Reduced Model
- 361 Intercomparison Project (RCMIP; Nicholls et al. 2020, 2021) are used. FaIR uses NOx emissions from the
- 362 aviation sector as a proxy for radiative forcing from contrails and contrail-induced cirrus, and this data is
- also infilled from the SSP scenarios using the RMS Closest method.
- In addition, the AR6 and SSP scenarios only run to 2100, whereas we require emissions to 2300. We
- follow the logic discussed in Meinshausen et al. (2020) to extend SSP emissions beyond 2100. If CO₂ FFI
- emissions are positive in 2100, they are linearly ramped down to zero by 2250; if they are negative, they
- are ramped up to zero by 2150. CO₂ AFOLU emissions are ramped to zero by 2150 whether negative or
- 368 positive in 2100. For CH₄, N₂O and short-lived forcers, the FFI component (always positive) is ramped
- down to zero by 2250, and the AFOLU component is held constant at 2100 emissions. As IAM scenario
- data does not provide the FFI/AFOLU split for non-CO₂ emissions, we estimate this from the average of
- the 2100 ratios in SSP scenarios that do have this granularity. Finally, minor greenhouse gases with no
- assumed AFOLU source are ramped down to their pre-industrial emissions level by 2250. Historical
- emissions from 1750-2014 are provided from RCMIP and for the period 2015-2020 we follow RFF-SP and
- 374 use SSP2-4.5.
- The final set of 10,000 infilled, complete RFF-SP emissions scenarios are then run using a 1001-member
- 376 probabilistic ensemble of a state-of-the-art simple climate model: the Finite amplitude Impulse
- 377 Response model (FaIR v2.1) (Leach et al. 2021). The FaIR calibration we use (v1.0; Smith 2023) is

- 378 consistent with the IPCC AR6 Working Group 1 assessment of present-day warming, equilibrium climate
- 379 sensitivity, transient climate response, present-day aerosol radiative forcing, present-day CO₂
- 380 concentrations, and recent-past ocean heat content change, including the uncertainties in these
- distributions (Forster et al. 2021; Smith et al. 2021), maintains a root-mean-square error between
- modelled and observed global mean surface temperature of less than 0.16°C in each ensemble member,
- and is consistent with historical emissions from RCMIP. We use stochastic, auto-correlated internal
- variability in temperature and forcing following Cummins et al. (2020). For comparison to the RFF-SP
- dataset, we also run the SSP scenarios under the same calibration for comparison. RCP scenarios also
- use this calibration but are additionally redrawn to historical emissions of short-lived forcers (Skeie et al.
- 387 2011).
- The coupled combination of 10,000 RFF-SP emissions scenarios, run with a 1001-member climate parameter ensemble, creates 10,010,000 total outcomes
- 390 (https://zenodo.org/record/7759089#.ZGI7kqXMK3C). Result data for each emissions scenario are saved
- 391 as a netCDF file and includes ensemble-level information about global mean surface temperature,
- 392 effective radiative forcing, ocean-heat content, and concentrations of CO₂, CH₄ and N₂O. Uncertainty
- 393 ranges are produced for temperature and radiative forcing and are defined by the following 3
- 394 classifications: "total uncertainty" represents the uncertainty range of all 10,010,000 outcomes,
- 395 "emissions uncertainty" represents the uncertainty range of median values for each climate parameter
- ensemble member, and "climate uncertainty" represents the uncertainty range associated with a single
- central emissions scenario. Uncertainty ranges are constrained by the 5th and 95th percentiles for the
- 398 years 2020 through 2300.
- 399

400 Code and data availability

- 401 RFF-SP emissions scenarios are available from https://doi.org/10.5281/zenodo.5898729.
- The AR6 Working Group 3 infilling database is available from https://doi.org/10.5281/zenodo.6390768
 (registration required).
- 404 RCMIP emissions prepared for the SSP scenarios are available from
- 405 <u>https://doi.org/10.5281/zenodo.4589756</u>.
- 406 FaIR v2.1 is available from https://doi.org/10.5281/zenodo.7459702 and v1.0 of the calibration is
- 407 available from (<u>https://doi.org/10.5281/zenodo.7545157</u>).
- The probabilistic FaIR output is available from https://zenodo.org/record/7838148 and code at https://github.com/chrisroadmap/rff-fair2.1.
- 410 The code for FaIR/Figure processing scripts is at
- 411 https://github.com/erysimumcap/RCM_RCP8.5_Probability
- 412
- 413 Acknowledgments

- 414 This research was funded by the U.S. Environmental Protection Agency under contract
- 415 #68HERH19D0027. The views expressed in this article are those of the authors and do not necessarily
- 416 represent the views or policies of the U.S. Environmental Protection Agency.
- 417

418 References

- 419 Ansolabehere et al. (2003), The Future of Nuclear Power, MIT, https://energy.mit.edu/wpcontent/uploads/2003/07/MITEI-The-Future-of-Nuclear-Power.pdf
- 420
- Barron, A. R. (2018). Time to refine key climate policy models. In Nature Climate Change (Vol. 8, Issue 5, 421 pp. 350–352). Springer Science and Business Media LLC. https://doi.org/10.1038/s41558-018-0132-y 422
- 423 Burgess, M. G., Ritchie, J., Shapland, J., & Pielke, R., Jr. (2020). IPCC baseline scenarios have over-
- 424 projected CO2 emissions and economic growth. In Environmental Research Letters (Vol. 16, Issue 1, p.
- 425 014016). IOP Publishing. https://doi.org/10.1088/1748-9326/abcdd2
- 426 Byers et al. 2022, Edward Byers, Volker Krey, Elmar Kriegler, Keywan Riahi, Roberto Schaeffer, Jarmo
- 427 Kikstra, Robin Lamboll, Zebedee Nicholls, Marit Sanstad, Chris Smith, Kaj-Ivar van der Wijst, Alaa Al
- 428 Khourdajie, Franck Lecocq, Joana Portugal-Pereira, Yamina Saheb, Anders Strømann, Harald Winkler,
- 429 Cornelia Auer, Elina Brutschin, Matthew Gidden, Philip Hackstock, Mathijs Harmsen, Daniel Huppmann,
- 430 Peter Kolp, Claire Lepault, Jared Lewis, Giacomo Marangoni, Eduardo Müller-Casseres, Ragnhild Skeie,
- 431 Michaela Werning, Katherine Calvin, Piers Forster, Celine Guivarch, Tomoko Hasegawa, Malte
- 432 Meinshausen, Glen Peters, Joeri Rogelj, Bjorn Samset, Julia Steinberger, Massimo Tavoni, Detlef van
- 433 Vuuren. AR6 Scenarios Database hosted by IIASA International Institute for Applied Systems Analysis,
- 434 2022. doi: 10.5281/zenodo.5886911 | url: data.ece.iiasa.ac.at/ar6/
- 435 Chen, D., M. Rojas, B.H. Samset, K. Cobb, A. Diongue Niang, P. Edwards, S. Emori, S.H. Faria, E. Hawkins,
- 436 P. Hope, P. Huybrechts, M. Meinshausen, S.K. Mustafa, G.-K. Plattner, and A.-M. Tréguier, 2021:
- 437 Framing, Context, and Methods. In Climate Change 2021: The Physical Science Basis. Contribution of
- 438 Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change
- 439 [Masson-Delmotte, V., P. Zhai, A. Pirani, S.L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb,
- 440 M.I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J.B.R. Matthews, T.K. Maycock, T. Waterfield, O. Yelekçi, R.
- 441 Yu, and B. Zhou (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA,
- 442 pp. 147-286, doi:10.1017/9781009157896.00
- 443 Christensen, P., Gillingham, K., & Nordhaus, W. (2018). Uncertainty in forecasts of long-run economic
- 444 growth. In Proceedings of the National Academy of Sciences (Vol. 115, Issue 21, pp. 5409–5414).
- 445 Proceedings of the National Academy of Sciences. <u>https://doi.org/10.1073/pnas.1713628115</u>
- 446 Cummins, D. P., Stephenson, D. B., & Stott, P. A. (2020). Optimal Estimation of Stochastic Energy Balance 447 Model Parameters. In Journal of Climate (Vol. 33, Issue 18, pp. 7909–7926). American Meteorological 448 Society. https://doi.org/10.1175/jcli-d-19-0589.1
- 449 Forster, P., T. Storelvmo, K. Armour, W. Collins, J.-L. Dufresne, D. Frame, D.J. Lunt, T. Mauritsen, M.D.
- 450 Palmer, M. Watanabe, M. Wild, and H. Zhang, 2021: The Earth's Energy Budget, Climate Feedbacks, and
- 451 Climate Sensitivity. In Climate Change 2021: The Physical Science Basis. Contribution of Working Group I
- 452 to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change [Masson-Delmotte,

- 453 V., P. Zhai, A. Pirani, S.L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M.I. Gomis, M.
- 454 Huang, K. Leitzell, E. Lonnoy, J.B.R. Matthews, T.K. Maycock, T. Waterfield, O. Yelekçi, R. Yu, and B. Zhou
- 455 (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, pp. 923–1054,
- 456 doi:<u>10.1017/9781009157896.009</u>.
- 457 Fredriksen, H-B., Smith, C.J., Modak, A., & Rugenstein, M. (2023). 21st Century Scenario Forcing
- 458 Increases More for CMIP6 Than CMIP5 Models. Geophysical Research Letters, Vol 50.
- 459 doi:10.1029/2023GL102916
- 460 Gidden, M. J., Fujimori, S., van den Berg, M., Klein, D., Smith, S. J., van Vuuren, D. P., & Riahi, K. (2018). A
- 461 methodology and implementation of automated emissions harmonization for use in Integrated
- 462 Assessment Models. In Environmental Modelling & Software (Vol. 105, pp. 187–200). Elsevier BV.
- 463 <u>https://doi.org/10.1016/j.envsoft.2018.04.002</u>
- 464 Hartin, C., McDuffie, E. E., Noiva, K., Sarofim, M., Parthum, B., Martinich, J., Barr, S., Neumann, J.,
- 465 Willwerth, J., and Fawcett, A.: Advancing the estimation of future climate impacts within the United
- 466 States, EGUsphere [preprint], https://doi.org/10.5194/egusphere-2023-114, 2023.
- Hausfather, Z., & Peters, G. P. (2020, January 29). Emissions the 'business as usual' story is misleading.
 Nature News. Retrieved May 7, 2023, from <u>https://www.nature.com/articles/d41586-020-00177-3</u>
- 469 Hawkins, E., & Sutton, R. (2009). The Potential to Narrow Uncertainty in Regional Climate Predictions. In
- 470 Bulletin of the American Meteorological Society (Vol. 90, Issue 8, pp. 1095–1108). American
- 471 Meteorological Society. <u>https://doi.org/10.1175/2009bams2607.1</u>
- 472 Huard, D., Fyke, J., Capellán-Pérez, I., Matthews, H. D., & Partanen, A. (2022). Estimating the Likelihood
- 473 of GHG Concentration Scenarios From Probabilistic Integrated Assessment Model Simulations. In Earth's
- 474 Future (Vol. 10, Issue 10). American Geophysical Union (AGU). <u>https://doi.org/10.1029/2022ef002715</u>
- 475 Kemp, L., Xu, C., Depledge, J., Ebi, K. L., Gibbins, G., Kohler, T. A., Rockström, J., Scheffer, M.,
- 476 Schellnhuber, H. J., Steffen, W., & Lenton, T. M. (2022). Climate Endgame: Exploring catastrophic climate
- 477 change scenarios. In Proceedings of the National Academy of Sciences (Vol. 119, Issue 34). Proceedings
- 478 of the National Academy of Sciences. <u>https://doi.org/10.1073/pnas.2108146119</u>
- 479 Kikstra, J. S., Nicholls, Z. R. J., Smith, C. J., Lewis, J., Lamboll, R. D., Byers, E., Sandstad, M., Meinshausen,
- 480 M., Gidden, M. J., Rogelj, J., Kriegler, E., Peters, G. P., Fuglestvedt, J. S., Skeie, R. B., Samset, B. H.,
- 481 Wienpahl, L., van Vuuren, D. P., van der Wijst, K.-I., Al Khourdajie, A., ... Riahi, K. (2022). The IPCC Sixth
- 482 Assessment Report WGIII climate assessment of mitigation pathways: from emissions to global
- temperatures. In Geoscientific Model Development (Vol. 15, Issue 24, pp. 9075–9109). Copernicus
- 484 GmbH. <u>https://doi.org/10.5194/gmd-15-9075-2022</u>
- Lamboll, R. D., Nicholls, Z. R. J., Kikstra, J. S., Meinshausen, M., & Rogelj, J. (2020). Silicone v1.0.0: an
- 486 open-source Python package for inferring missing emissions data for climate change research. In
- 487 Geoscientific Model Development (Vol. 13, Issue 11, pp. 5259–5275). Copernicus GmbH.
- 488 <u>https://doi.org/10.5194/gmd-13-5259-2020</u>
- 489 Leach, N. J., Jenkins, S., Nicholls, Z., Smith, C. J., Lynch, J., Cain, M., Walsh, T., Wu, B., Tsutsui, J., & Allen,
- 490 M. R. (2021). FalRv2.0.0: a generalized impulse response model for climate uncertainty and future

- 491 scenario exploration. In Geoscientific Model Development (Vol. 14, Issue 5, pp. 3007–3036). Copernicus
 492 GmbH. <u>https://doi.org/10.5194/gmd-14-3007-2021</u>
- 493 Lehner, F., Deser, C., Maher, N., Marotzke, J., Fischer, E. M., Brunner, L., Knutti, R., & Hawkins, E. (2020).
- 494 Partitioning climate projection uncertainty with multiple large ensembles and CMIP5/6. In Earth System
- 495 Dynamics (Vol. 11, Issue 2, pp. 491–508). Copernicus GmbH. <u>https://doi.org/10.5194/esd-11-491-2020</u>
- Liu, P. R., & Raftery, A. E. (2021). Country-based rate of emissions reductions should increase by 80%
- 497 beyond nationally determined contributions to meet the 2 °C target. In Communications Earth &
- 498 Environment (Vol. 2, Issue 1). Springer Science and Business Media LLC. <u>https://doi.org/10.1038/s43247-</u>
 499 021-00097-8
- 500 Meinshausen, M., Nicholls, Z. R. J., Lewis, J., Gidden, M. J., Vogel, E., Freund, M., Beyerle, U., Gessner, C.,
- 501 Nauels, A., Bauer, N., Canadell, J. G., Daniel, J. S., John, A., Krummel, P. B., Luderer, G., Meinshausen, N.,
- 502 Montzka, S. A., Rayner, P. J., Reimann, S., ... Wang, R. H. J. (2020). The shared socio-economic pathway
- 503 (SSP) greenhouse gas concentrations and their extensions to 2500. In Geoscientific Model Development
- 504 (Vol. 13, Issue 8, pp. 3571–3605). Copernicus GmbH. <u>https://doi.org/10.5194/gmd-13-3571-2020</u>
- Morris, J., J. Reilly, S. Paltsev, A. Sokolov, and K Cox (2022), Representing Socio-Economic Uncertainty in
 Human System Models. Earth's Future, <u>https://doi.org/10.1029/2021EF002239</u>
- 507 Moss, R. H., J. A. Edmonds, K. A. Hibbard, M. R. Manning, S. K. Rose, D. P. van Vuuren, T. R. Carter, S.
- 508 Emori, M. Kainuma, T. Kram, G. A. Meehl, J. F. B. Mitchell, N. Nakicenovic, K. Riahi, S. J. Smith, R. J.
- 509 Stouffer, A. M. Thomson, J. P. Weyant, and T. J. Wilbanks, 2010: The next generation of scenarios for
- 510 climate change research and assessment. Nature, 463, 747–756. doi:10.1038/nature08823
- 511 Nicholls, Z. R. J., Meinshausen, M., Lewis, J., Gieseke, R., Dommenget, D., Dorheim, K., Fan, C.-S.,
- 512 Fuglestvedt, J. S., Gasser, T., Golüke, U., Goodwin, P., Hartin, C., Hope, A. P., Kriegler, E., Leach, N. J.,
- 513 Marchegiani, D., McBride, L. A., Quilcaille, Y., Rogelj, J., ... Xie, Z. (2020). Reduced Complexity Model
- 514 Intercomparison Project Phase 1: introduction and evaluation of global-mean temperature response. In
- 515 Geoscientific Model Development (Vol. 13, Issue 11, pp. 5175–5190). Copernicus GmbH.
- 516 https://doi.org/10.5194/gmd-13-5175-2020
- 517 Nicholls, Z., Meinshausen, M., Lewis, J., Corradi, M. R., Dorheim, K., Gasser, T., Gieseke, R., Hope, A. P.,
- Leach, N. J., McBride, L. A., Quilcaille, Y., Rogelj, J., Salawitch, R. J., Samset, B. H., Sandstad, M.,
- 519 Shiklomanov, A., Skeie, R. B., Smith, C. J., Smith, S. J., ... Woodard, D. L. (2021). Reduced Complexity
- 520 Model Intercomparison Project Phase 2: Synthesizing Earth System Knowledge for Probabilistic Climate
- 521 Projections. In Earth's Future (Vol. 9, Issue 6). American Geophysical Union (AGU).
- 522 <u>https://doi.org/10.1029/2020ef001900</u>
- 523 O'Neill, B. C., Tebaldi, C., van Vuuren, D. P., Eyring, V., Friedlingstein, P., Hurtt, G., Knutti, R., Kriegler, E.,
- Lamarque, J.-F., Lowe, J., Meehl, G. A., Moss, R., Riahi, K., & Sanderson, B. M. (2016). The Scenario
- 525 Model Intercomparison Project (ScenarioMIP) for CMIP6. In Geoscientific Model Development (Vol. 9,
- 526 Issue 9, pp. 3461–3482). Copernicus GmbH. <u>https://doi.org/10.5194/gmd-9-3461-2016</u>
- 527 Pielke Jr, R., Burgess, M. G., & Ritchie, J. (2022). Plausible 2005–2050 emissions scenarios project
- 528 between 2 °C and 3 °C of warming by 2100. In Environmental Research Letters (Vol. 17, Issue 2, p.
- 529 024027). IOP Publishing. <u>https://doi.org/10.1088/1748-9326/ac4ebf</u>

- Rennert, K. et al. (2022a) The social cost of carbon: advances in long-term probabilistic projections of
 population, GDP, emissions, and discount rates. Brook. Pap. Econ. Act. Fall 2021, 223–275.
- Rennert, K., Errickson, F., Prest, B. C., Rennels, L., Newell, R. G., Pizer, W., Kingdon, C., Wingenroth, J.,
- 533 Cooke, R., Parthum, B., Smith, D., Cromar, K., Diaz, D., Moore, F. C., Müller, U. K., Plevin, R. J., Raftery, A.
- 534 E., Ševčíková, H., Sheets, H., ... Anthoff, D. (2022). Comprehensive evidence implies a higher social cost
- of CO2. In Nature (Vol. 610, Issue 7933, pp. 687–692). Springer Science and Business Media LLC.
- 536 https://doi.org/10.1038/s41586-022-05224-9
- 537 Riahi, K., Rao, S., Krey, V., Cho, C., Chirkov, V., Fischer, G., Kindermann, G., Nakicenovic, N., & Rafaj, P.
- 538 (2011). RCP 8.5—A scenario of comparatively high greenhouse gas emissions. In Climatic Change (Vol.
- 539 109, Issues 1–2, pp. 33–57). Springer Science and Business Media LLC. <u>https://doi.org/10.1007/s10584-</u>
- 540 <u>011-0149-y</u>
- 541 Riahi Keywan, Detlef P. van Vuuren, Elmar Kriegler, Jae Edmonds, Brian C. O'Neill, Shinichiro Fujimori,
- 542 Nico Bauer, Katherine Calvin, Rob Dellink, Oliver Fricko, Wolfgang Lutz, Alexander Popp, Jesus Crespo
- 543 Cuaresma, Samir KC, Marian Leimbach, Leiwen Jiang, Tom Kram, Shilpa Rao, Johannes Emmerling, Kristie
- 544 Ebi, Tomoko Hasegawa, Petr Havlík, Florian Humpenöder, Lara Aleluia Da Silva, Steve Smith, Elke
- 545 Stehfest, Valentina Bosetti, Jiyong Eom, David Gernaat, Toshihiko Masui, Joeri Rogelj, Jessica Strefler,
- 546 Laurent Drouet, Volker Krey, Gunnar Luderer, Mathijs Harmsen, Kiyoshi Takahashi, Lavinia Baumstark,
- 547 Jonathan C. Doelman, Mikiko Kainuma, Zbigniew Klimont, Giacomo Marangoni, Hermann Lotze-Campen,
- 548 Michael Obersteiner, Andrzej Tabeau, Massimo Tavoni. The Shared Socioeconomic Pathways and their
- energy, land use, and greenhouse gas emissions implications: An overview, Global Environmental
- 550 Change, Volume 42, Pages 153-168, 2017, ISSN 0959-3780, DOI:110.1016/j.gloenvcha.2016.05.009
- 551 Riahi, K., R. Schaeffer, J. Arango, K. Calvin, C. Guivarch, T. Hasegawa, K. Jiang, E. Kriegler, R. Matthews,
- 552 G.P. Peters, A. Rao, S. Robertson, A.M. Sebbit, J. Steinberger, M. Tavoni, D.P. van Vuuren, 2022:
- 553 Mitigation pathways compatible with long-term goals. In IPCC, 2022: Climate Change 2022: Mitigation of
- 554 Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the
- 555 Intergovernmental Panel on Climate Change [P.R. Shukla, J. Skea, R. Slade, A. Al Khourdajie, R. van
- 556 Diemen, D. McCollum, M. Pathak, S. Some, P. Vyas, R. Fradera, M. Belkacemi, A. Hasija, G. Lisboa, S. Luz,
- 557 J. Malley, (eds.)]. Cambridge University Press, Cambridge, UK and New York, NY, USA.
- 558 doi:10.1017/9781009157926.005
- 559 Rogelj, J., Rao, S., McCollum, D. L., Pachauri, S., Klimont, Z., Krey, V., & Riahi, K. (2014). Air-pollution
- 560 emission ranges consistent with the representative concentration pathways. In Nature Climate Change
- 561 (Vol. 4, Issue 6, pp. 446–450). Springer Science and Business Media LLC.
- 562 <u>https://doi.org/10.1038/nclimate2178</u>
- 563 Sarofim, M. C., Martinich, J., Neumann, J. E., Willwerth, J., Kerrich, Z., Kolian, M., Fant, C., & Hartin, C.
- 564 (2021). A temperature binning approach for multi-sector climate impact analysis. In Climatic Change
- 565 (Vol. 165, Issues 1–2). Springer Science and Business Media LLC. <u>https://doi.org/10.1007/s10584-021-</u>
- 566 <u>03048-6</u>
- 567 Schleussner, C.-F., Lissner, T. K., Fischer, E. M., Wohland, J., Perrette, M., Golly, A., Rogelj, J., Childers, K.,
- 568 Schewe, J., Frieler, K., Mengel, M., Hare, W., & Schaeffer, M. (2016). Differential climate impacts for

- policy-relevant limits to global warming: the case of 1.5 °C and 2 °C. In Earth System Dynamics (Vol. 7,
- 570 Issue 2, pp. 327–351). Copernicus GmbH. <u>https://doi.org/10.5194/esd-7-327-2016</u>
- 571 Schwalm, C. R., Glendon, S., & Duffy, P. B. (2020). RCP8.5 tracks cumulative CO2 emissions. In
- 572 Proceedings of the National Academy of Sciences (Vol. 117, Issue 33, pp. 19656–19657). Proceedings of
- 573 the National Academy of Sciences. <u>https://doi.org/10.1073/pnas.2007117117</u>
- 574 Skea, J., van Diemen, R., Portugal-Pereira, J., & Khourdajie, A. A. (2021). Outlooks, explorations and
- 575 normative scenarios: Approaches to global energy futures compared. In Technological Forecasting and
- 576 Social Change (Vol. 168, p. 120736). Elsevier BV. <u>https://doi.org/10.1016/j.techfore.2021.120736</u>
- 577 Skeie, R. B., Berntsen, T. K., Myhre, G., Tanaka, K., Kvalevåg, M. M., & Hoyle, C. R. (2011). Anthropogenic
- 578 radiative forcing time series from pre-industrial times until 2010. In Atmospheric Chemistry and Physics
- 579 (Vol. 11, Issue 22, pp. 11827–11857). Copernicus GmbH. <u>https://doi.org/10.5194/acp-11-11827-2011</u>
- 580 Smith, C., Z.R.J. Nicholls, K. Armour, W. Collins, P. Forster, M. Meinshausen, M.D. Palmer, and M.
- 581 Watanabe, 2021: The Earth's Energy Budget, Climate Feedbacks, and Climate Sensitivity Supplementary
- 582 Material. In Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the
- 583 Sixth Assessment Report of the Intergovernmental Panel on Climate Change [Masson-Delmotte, V., P.
- 584 Zhai, A. Pirani, S.L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M.I. Gomis, M. Huang, K.
- Leitzell, E. Lonnoy, J.B.R. Matthews, T.K. Maycock, T. Waterfield, O. Yelekçi, R. Yu, and B. Zhou (eds.)].
- 586 Available from <u>https://www.ipcc.ch/</u>
- 587 Srikrishnan, V., Guan, Y., Tol, R. S. J., & Keller, K. (2022). Probabilistic projections of baseline twenty-first
- 588 century CO2 emissions using a simple calibrated integrated assessment model. In Climatic Change (Vol.
- 589 170, Issues 3–4). Springer Science and Business Media LLC. <u>https://doi.org/10.1007/s10584-021-03279-7</u>
- Tebaldi, C., Debeire, K., Eyring, V., Fischer, E., Fyfe, J., Friedlingstein, P., Knutti, R., Lowe, J., O'Neill, B.,
- 591 Sanderson, B., van Vuuren, D., Riahi, K., Meinshausen, M., Nicholls, Z., Tokarska, K. B., Hurtt, G., Kriegler,
- 592 E., Lamarque, J.-F., Meehl, G., ... Ziehn, T. (2021). Climate model projections from the Scenario Model
- 593 Intercomparison Project (ScenarioMIP) of CMIP6. In Earth System Dynamics (Vol. 12, Issue 1, pp. 253–
- 594 293). Copernicus GmbH. <u>https://doi.org/10.5194/esd-12-253-2021</u>
- 595 USGCRP, 2015, U.S. Global Change Research Program General Decisions Regarding Climate-Related
- 596 Scenarios for Framing the Fourth National Climate Assessment,
- 597 <u>https://scenarios.globalchange.gov/sites/default/files/External%20memo%20NCA4%20scenarios%20fra</u>
 598 <u>ming_20150506.pdf</u>
- 599 Sokolov, A. P., Stone, P. H., Forest, C. E., Prinn, R., Sarofim, M. C., Webster, M., Paltsev, S., Schlosser, C.
- A., Kicklighter, D., Dutkiewicz, S., Reilly, J., Wang, C., Felzer, B., Melillo, J. M., & Jacoby, H. D. (2009).
- 601 Probabilistic Forecast for Twenty-First-Century Climate Based on Uncertainties in Emissions (Without
- Policy) and Climate Parameters. In Journal of Climate (Vol. 22, Issue 19, pp. 5175–5204). American
- 603 Meteorological Society. <u>https://doi.org/10.1175/2009jcli2863.1</u>
- 604 Wyser, K., Kjellström, E., Koenigk, T., Martins, H., & Döscher, R. (2020). Warmer climate projections in
- 605 EC-Earth3-Veg: the role of changes in the greenhouse gas concentrations from CMIP5 to CMIP6. In
- 606 Environmental Research Letters (Vol. 15, Issue 5, p. 054020). IOP Publishing.
- 607 https://doi.org/10.1088/1748-9326/ab81c2